Author's response to reviews

Title: Helicobacter pylori Infection and its Association with Anemia among Adult Dyspeptic Patients Attending Butajira Hospital, Ethiopia

Authors:

Dargaze Kibru (kdargaze@yahoo.com)
Baye Gelaw (tedybayegelaw@gmail.com)
Agersew Alemu (agersewalemu@yahoo.com)
Zelalem Addis (zelalemaddis@ymail.com)

Version: 5  Date: 13 November 2014

Author's response to reviews: see over
Dear editors,

Thank you very much for accepting our manuscript to be published. As per your comment we made the necessary revision.

Informed written consent was obtained from study participants and this is indicated in line 131 and 132 under the ethical consideration part. As indicated the study included only adult patients so children were excluded and no consent form was needed.